Author's response to reviews

Title: Fluid Overload in the ICU: Evaluation and Management

Authors:

Rolando Claure-Del Granado (rclaure@yahoo.com)
Ravindra Mehta (rmehta@ucsd.edu)

Version: 2 Date: 27 Jun 2016

Author's response to reviews:

Dear Rosemary Philpott,

BMC Nephrology

Thank you for considering our manuscript "Fluid Overload in the ICU: Evaluation and Management" (BNEP-D-15-00076R1) for potentially being acceptable for publication in BMC Nephrology. We have addressed the following points:

1. We note that the current submission contains some textual overlap with other previously published works. Please refer to the attached iThenticate document and re-phrase these sections in your own words to ensure no overlap.

   - All the sections with some textual overlap with previously published work have been re-phrased.

2. Please change the heading 'Summary' to 'Conclusions'.

   - We have changed the heading as requested
3. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

- A clean copy of the manuscript was uploaded

4. Please add a ' Declarations' heading to your manuscript before the references section. Please ensure that this section contains all of the following sub-sections:

- A "Declaration" heading was added, and the following sub-sections that were missing on the previous submission were provided as requested: List of abbreviations, ethics and consent to participate, Acknowledgements, consent to publish, and availability of data and materials.